Author's response to reviews

Title: Common sequence variants in the LOXL1 gene in pigment dispersion syndrome and pigmentary glaucoma

Authors:

EMILIANO GIARDINA (giardina@uniroma2.it)
FRANCESCO ODDONE (oddonef@gmail.com)
TIZIANA LEPRE (tiziana.lepre@gmail.com)
MARCO CENTOFANTI (marco.centofanti@fastwebnet.it)
CRISTINA PECONI (cristina.peconi@gmail.com)
LUCIA TANGA (lucia.tanga@gmail.com)
LUCIANO QUARANTA (quaranta@med.unibs.it)
PAOLO FREZZOTTI (paolo.frezzotti@gmail.com)
GIUSEPPE NOVELLI (novelli@med.uniroma2.it)
GIANLUCA MANNI (glmanni53@gmail.com)

Version: 2 Date: 23 July 2013

Author's response to reviews:

Editors
BMC Ophthalmology

Dear Editor,

please find attached the manuscript entitled “Common sequence variants in the LOXL1 gene in pigment dispersion syndrome and pigmentary glaucoma” submitted to your journal for review and if suitable for publication. According to yours specification I added in the manuscript:

1. “The study was approved by the local ethics committee (Ethics Committee of the University of Tor Vergata, R. S. 91.10, 24/09/2010) and adhered to the tenets of the Declaration of Helsinki and local law” in the methods section.

2. “Competing interests: The authors declare that they have no competing interests”.

3. Authors' contributions

EG, FO, MC, LT, GN and GM designed the study. LT, FO, MC collected data. TL, CP carried out the laboratory procedures. FO, MC, LT performed data analysis. EG, FO drafted the manuscript. All authors revised the manuscript. All authors read and approved the final manuscript.

Hoping you may find our paper interesting I thank you for considering it

Best regards
Lucia Tanga, MD.
Via Livenza 3
00198 Rome, Italy
Tel: +39-(0)6-85356727
Email: lucia.tanga@gmail.com